Author’s response to reviews

Title: Awareness of vaccination status and its predictors among working people in Switzerland

Authors:

Chung-Yol Lee (chung-yol.lee@bag.admin.ch)
Claudine Naguel (nac@meteoswiss.ch)
Danielle Gyurech (reisemedizin@compuserve.com)
Nicole Duvoisin (nicoled@ifspm.unizh.ch)
Dr Julian Schilling (juli@ifspm.unizh.ch)

Version: 3 Date: 19 May 2003

Reply to Referee PA Bovier

Thank you very much for reviewing our paper once again.

We have now made all the changes you suggested. We have omitted Table 2 and simplified Table 3 by presenting only the significant factors. There is now only one mention per Table and Figure in the results section.

Thank you for drawing our attention to the 95% confidence intervals, they are all below 1%. These intervals have now been included in Figure 1. A mistake was made in the Chi squared test formula performed with Excel. The proportions are indeed statistically different when performing the chi squared test (p<0.001) as well as for the SPSS (also p < 0.001) Cochrane test.

Thank you for your time and consideration.

Kind regards

Julian Schilling